Author's response to reviews

Title: Improved cardiovascular diagnostic accuracy by pocket size imaging device in non cardiologic outpatients. The NaUSiCa (Naples Ultrasound Stethoscope in Cardiology) Study

Authors:

Maurizio Galderisi (mgalderi@unina.it)
Alessandro Santoro (cantopo@hotmail.com)
Marco Versiero (mailpermarco@tiscali.it)
Vincenzo Schiano Lomoriello (capatost@libero.it)
Roberta Esposito (robyeire@tin.it)
Rosa Raia (rosa.78@libero.it)
Francesca Farina (francesca.farina81@gmail.com)
Pier Luigi Schiattarella (pluigi@libero.it)
Manuela Bonito (manu.bonito@hotmail.it)
Marinella Olibet (marinellaolibet@libero.it)
Giovanni de Simone (simogi@unina.it)

Version: 3 Date: 19 November 2010

Author's response to reviews:

Notes for the Editor:
The Abstracts in the submissions system and manuscript are identical
Authors' contributions: have been stated at the end of the manuscript text, directly preceding the reference list
`Competing interests' section have been included between the Conclusions and Authors' contributions. `The authors declare that they have no competing interests'.
Additional files: have been removed..
The Tables have been labeled appropriately and sequentially.
Title written according to the editorial rules.
All colored texts are now black
The 'authors list', and 'emails' have been formatted into separate lists.
All the highlighting have been removed from the manuscript.
The Keywords have been removed.
The visible vertical lines have been removed from the tables